\begin{table}%t1
\caption{Information about the quasar-galaxy pairs in our
sample. $z_{\rm Q}$ is the redshift of the quasar and $z_{\rm G}$ is the redshift
of the galaxy. The last two columns list what filters were used in our
analysis and the references for the magnitudes.}
\label{table:info}
\small%\centerline
{
\begin{tabular} {l l c c l l l}
\hline \hline
Quasar & Images & $z_{\rm Q}$ & $z_{\rm G}$ & Galaxy type & Filters & Source$^a$ \\
\hline
SDSS J131058.13+010822.2             & A            &  1.39  &  0.04  & star forming & {\it u g r i z }       &             1         \\
Q2237+030                            & A B C D      &  1.69  &  0.04  & late &{\it  F160W F205W F555W F675W F814W} &   2          \\
SDSS J114719.89+522923.1             & A            &  1.99  &  0.05  & starburst & {\it u g r i z }                 &  1          \\
SDSS J084957.97+510829.0             & A            &  0.58  &  0.07  & starburst & {\it u g r i z }              &      1          \\
SDSS1155+6346                        & A B          &  2.89  &  0.18  & early & {\it F160W F555W F814W K } & 2, 3            \\
MG1654+1346                          & A            &  1.74  &  0.25  & elliptical & {\it F160W F555W F675W F814W}& 2            \\
CXOCY J220132.8-320144               & A B          &  3.90  &  0.32  & spiral & {\it g r i Ks }               &     4           \\
SDSS J0903+5028                      & A B          &  3.61  &  0.39  & early & {\it F160W F555W F814W r i }  & 2, 5               \\
SDSS J0924+0219                      & A B C        &  1.52  &  0.39  & elliptical & {\it u g r i F160W F555W F814W} &2, 6            \\
CLASS B1152+199                      & A B          &  1.02  &  0.44  & late & {\it F160W F555W F814W I R V} & 2, 7       \\
HE0435-1223                          & A B C D      &  1.69  &  0.45  & S0 & {\it F160W F555W F814W i g r}&  2, 8            \\
Q0142-100                            & A B          &  2.72  &  0.49  & early & {\it F160W F555W F675W F814W}&  2            \\
HE0230-2130                          & A1 A2 B C    &  2.16  &  0.52  & S0/Sa & {\it B R I K F814W F555W }   &  2, 9              \\
BRI0952-0115                         & A B          &  4.43  &  0.63  & elliptical & {\it F160W F555W F675W F814W}   &2          \\
WFI2033-4723                         & A1 A2 B C    &  1.66  &  0.66  & Sb/Sc& {\it F160W F555W F814W i}   &    2, 10  \\
SDSS J1004+4112                      & A B C D      &  1.74  &  0.68  & cluster of galaxies&{\it  u g r i z F160W F555W F814W} & 2, 11     \\
J1004+1229                           & A B          &  2.65  &  0.95  & elliptical & {\it F160W F555W F814W I }        &  2  \\
MG0414+0534                          & A1 A2 B C    &  2.64  &  0.96  & early &{\it  F160W F110W F205W F675W F814W} & 2 \\
SDSS J012147.73+002718.7             & A            &  2.22  &  1.39  &unknown& {\it u g r i z }    &       1 \\
SDSS J145907.19+002401.2             & A            &  3.01  &  1.39  &unknown& {\it u g r i z }     &      1 \\
SDSS J144612.98+035154.4             & A            &  1.95  &  1.51  &unknown& {\it u g r i z}     &       1 \\
\hline
\end{tabular}}
\smallskip
$^a$  1: {\citet{sdss}}; 2: {\citet{castles}}; 3: {\citet{pindor04}}; 4: {\citet{castander06}}; 5:~{\citet{johnston03}}; 6: {\citet{inada03}}; 7: {\citet{toft00}}; 8: {\citet{wisotzki02}}; 9: {\citet{wisotzki99}}; 10: {\citet{morgan04}}; 11:~{\citet{oguri04}}.  
\end{table}